Author’s response to reviews

Title: GLUT1 gene is a potential hypoxic and prognostic marker in colorectal cancer patients

Authors:

Fu Yen Chung (fuyenchung@gmail.com)
Ming Yii Huang (miyihu@kmu.edu.tw)
Ching Sheng Yeh (janson.yeh@msa.hinet.net)
Tian Lu Cheng (tlcheng@kmu.edu.tw)
Li Chen Yen (lcyen.fy@gmail.com)
Jaw Yuan Wang (cy614112@ms14.hinet.net)
Shiu Ru Lin (srlin@ms2.hinet.net)

Version: 3  Date: 5 February 2009

Author’s response to reviews:

Dear Editor-in-Chief:

I am very grateful to you, and all of the reviewers for paying such close attention, going over the entire paper in such detail, and providing valuable suggestions. In the second revised manuscript, I have answered the red text and the parts marked for corrections in greater detail. I have also answered the questions from each reviewer in the cover letter. As for the errors in English, the entire document, with the reviewers’ comments, were forwarded to one of the laboratory’s foreign professional English instructors for correction. I hope that doing so will make the manuscript conform to the requirements of BMC Cancer. I believed that the problems encountered in the first revised paper occurred partly because I was in a rush to attend a seminar abroad and did not have enough time for detailed communication with the young researcher who was responsible for the revision and partly because our laboratory’s foreign professional English instructor did not thoroughly double check it before submission. I feel quite sorry for causing you so much trouble, and I appreciate your understanding and assistance in this matter.

In regard to the reviewers’ comments, detailed explanations and replies are as follows:

Author’s response to reviews
Title: GLUT1 Gene is a Potential Hypoxic and Prognostic Marker in Colorectal Cancer Patients
Authors:
#FYC: fuyenchung@gmail.com
#MYH: miyihu@kmu.edu.tw
CSY: janson.yeh@msa.hinet.net